Author’s response to reviews

Title: Asthma and atopy prevalence are not reduced among former tuberculosis patients compared with controls in Lima, Peru

Authors:

Anthony Byrne (abyrne@med.usyd.edu.au)
Ben Marais (ben.marais@sydney.edu.au)
Carole Mitnick (carole_mitnick@hms.harvard.edu)
Frances Garden (frances.garden@sydney.edu.au)
Leonid Lecca (llecca_ses@pih.org)
Carmen Contreras (Ccontreras_SES@pih.org)
Yaninna Yauri (yanihj@yahoo.es)
Fanny Garcia (fgarcia_ses@pih.org)
Guy Marks (g.marks@unsw.edu.au)

Version: 2 Date: 29 Jan 2019

Author’s response to reviews:

Dear Mr Hickey,

thank you for the opportunity to revise our manuscript. Please find our responses to your four queries below;

Query 1.
Please state whether the maps depicted in figure 1 are your own or taken from another source.

Response:
Figure 1 is our own.
Query 2.

Please remove any figure titles and captions from the figure files as these files should contain the image graphics only. Upon doing this, please place the titles/captions at the end of the main manuscript after the References section under a newly created “Figure legend” heading.

Response:

This has been done now. Thank you.

Query 3.

Please remove the STROBE checklist and the response to reviewers from the file inventory, as they are no longer needed at this stage of the editorial process.

Response:

Now done, thank you.

Query 4.

When submitting your revised manuscript please ensure you do so as a single clean copy without any tracked changes, colored or highlighted text, as these are no longer required at this stage of the editorial process.

Response:

We have submitted a clean copy as requested.

Kind regards,

Dr Anthony Byrne and co-authors